

Citation: Laynez-Roldán P, Losada-Galván I, Posada E, de la Torre Ávila L, Casellas A, Sanz S, et al. (2023) Characterization of Latin American migrants at risk for *Trypanosoma cruzi* infection in a non-endemic setting. Insights into initial evaluation of cardiac and digestive involvement. PLoS Negl Trop Dis 17(7): e0011330. https://doi. org/10.1371/journal.pntd.0011330

Editor: Richard Reithinger, RTI International, UNITED STATES

Received: October 7, 2022

Accepted: April 24, 2023

Published: July 13, 2023

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0011330

Copyright: © 2023 Laynez-Roldán et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data present in this study are available upon request from the

RESEARCH ARTICLE

Characterization of Latin American migrants at risk for *Trypanosoma cruzi* infection in a non-endemic setting. Insights into initial evaluation of cardiac and digestive involvement

Pedro Laynez-Roldán¹[®]*, Irene Losada-Galván^{1,2®}*, Elizabeth Posada¹, Leonardo de la Torre Ávila¹, Aina Casellas¹, Sergi Sanz¹, Carme Subirà¹, Natalia Rodriguez-Valero¹, Daniel Camprubí-Ferrer¹, Isabel Vera¹, Montserrat Roldán¹, Edelweiss Aldasoro³, Inés Oliveira-Souto⁴, Antonia Calvo-Cano⁵, Maria-Eugenia Valls^{1,6}, Míriam J. Álvarez-Martínez^{1,6}, Montserrat Gállego^{1,7,8}, Alba Abras⁹, Cristina Ballart^{1,7}, José Muñoz¹, Joaquim Gascón^{1,8}, Maria-Jesus Pinazo^{1,8}*

1 Barcelona Institute for Global Health, ISGlobal-Hospital Clinic, Universitat de Barcelona, Barcelona, Spain, 2 Hospital Universitario 12 de Octubre, Madrid, Spain, 3 The International Foundation for Integrated Care (IFIC), Schiphol, The Netherlands, 4 Department of Infectious Diseases, Tropical Medicine & International Health Unit Vall d'Hebron-Drassanes, Vall d'Hebron University Hospital, Universitat Autònoma de Barcelona, PROSICS Barcelona, Barcelona, Spain, 5 Department of Infectious Pathology, Badajoz University Hospital, Badajoz, Spain, 6 Department of Microbiology, Hospital Clinic, Barcelona, Spain, 7 Parasitology section, Departament of Biology, Health and Environment, Faculty of Pharmacy and Food Science, Universitat de Barcelona, Barcelona, Spain, 8 Center for Network Biothecnological Research in Infectious Diseases (CIBERISCIII), Madridy, Spain, 9 Genetic Area, Department of Biology, Universitat de Girona, Campus Montilivi, Girona, Spain

• These authors contributed equally to this work.

* playnezroldan@gmail.com (PL-R); irene.losada@isglobal.org (IL-G); mpinazo@dndi.org (M-JP)

Abstract

Background

Trypanosoma cruzi causes Chagas disease (CD), a potentially fatal disease characterized by cardiac disorders and digestive, neurological or mixed alterations. *T. cruzi* is transmitted to humans by the bite of triatomine vectors; both the parasite and disease are endemic in Latin America and the United States. In the last decades, population migration has changed the classic epidemiology of *T. cruzi*, contributing to its global spread to traditionally non-endemic countries. Screening is recommended for Latin American populations residing in non-endemic countries.

Methods

The present study analyzes the epidemiological characteristics of 2,820 Latin American individuals who attended the International Health Service (IHS) of the Hospital Clinic de Barcelona between 2002 and 2019. The initial assessment of organ damage among positive cases of *T. cruzi* infection was analyzed, including the results of electrocardiogram (ECG), echocardiogram, barium enema and esophagogram. "Catalan Open Research Area" with the identifier https://doi.org/10.34810/data223 (contact address: ubioesdm@isglobal.org), access to the data is restricted due to ethical reasons. The complete data dictionary (data_dictionary.pdf) listing the dataset variables, and the dataset details (readme. txt) are available without restriction. Additionally, the publication includes a supporting information file to facilitate reproducibility of Fig 1 (S1 Table).

Funding: ISGlobal work is supported by the Departament d'Universitats i Recerca de la Generalitat de Catalunya, Spain (AGAUR; 2017SGR00924) and by the Instituto de Salud Carlos III (ISCIII) RICET Network for Cooperative Research in Tropical Diseases (ISCIII; RD16/0027/ 0004 - PI1290) and FEDER. MJP research is supported by the Ministry of Health, Government of Catalonia (PERIS 2016-2010 SLT008/18/ 00132). The research of MJP and JG is supported by the Departament d'Universitats i Recerca de la Generalitat de Catalunya, Spain (AGAUR; grant 2014SGR26). ILG research is supported by "Centro de Excelencia Severo Ochoa 2019-2023" Program (CEX2018-000806-S), from the Spanish Ministry of Science and Innovation and State Research Agency. Specific funding has not been received for this study. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Results

Among all the screened individuals attending the clinic, 2,441 (86.6%) were born in Bolivia and 1,993 (70.7%) were female. Of individuals, 1,517 (81.5%) reported previous exposure to the vector, which is a strong risk factor associated with *T. cruzi* infection; 1,382 individuals were positive for *T. cruzi* infection. The first evaluation of individuals with confirmed *T. cruzi* infection, showed 148 (17.1%) individuals with Chagasic cardiomyopathy, the main diagnostic method being an ECG and the right bundle branch block (RBBB) for the most frequent disorder; 16 (10.8%) individuals had a normal ECG and were diagnosed of Chagasic cardiomyopathy by echocardiogram.

Conclusions

We still observe many Latin American individuals who were at risk of *T. cruzi* infection in highly endemic areas in their countries of origin, and who have not been previously tested for *T. cruzi* infection. In fact, even in Spain, a country with one of the highest proportion of diagnosis of Latin American populations, *T. cruzi* infection remains underdiagnosed. The screening of Latin American populations presenting with a similar profile as reported here should be promoted. ECG is considered necessary to assess Chagasic cardiomyopathy in positive individuals, but echocardiograms should also be considered as a diagnostic approach given that it can detect cardiac abnormalities when the ECG is normal.

Author summary

Trypanosoma cruzi is a protozoan infection that can be transmitted to humans by triatomine insects, endemic from 21 Latin American countries. It can also be transmitted vertically (mother to child) and by blood transfusions, among other less frequent methods. T. cruzi infection is called Chagas Disease (CD) when causing organ damage, such as Chagasic cardiomyopathy (CC) and digestive involvement in 30-40% of cases. The large migration flows from Latin America to Europe have globalized the distribution of *T. cruzi* infection. Thus, screening is recommended for Latin Americans living in non-endemic countries. We have retrospectively analyzed the epidemiologic characteristics of individuals from endemic countries screened for T. cruzi infection in Barcelona over 17 years, revealing a great number of working-age women coming from highly endemic areas for T. cruzi infection, which reflect the migration movements of the last decades and help us to focus the screening and health promotion programs. We have also analyzed the initial organ damage assessment, which revealed a great proportion of right bundle branch block and left anterior fascicular block, considered typical CD lesions (although unspecific). We also found an important proportion of patients with an altered echocardiogram but having a normal ECG, which reinforces the echocardiogram as an essential test for the assessment of CC.

Introduction

American trypanosomiasis was first described by Carlos Chagas in 1909 [1]. Currently, we refer to *Trypanosoma cruzi* infection when no evidence of organ damage is present, while Chagas disease (CD) is the term used for the illness that results from *T. cruzi* infection. Up to 21

Latin American countries and the United States are considered endemic for *T. cruzi* infection and it is estimated to affect 6–7 million people [2]. Of all endemic countries, Bolivia has by far the highest reported prevalence, currently estimated at 6.1% [2]. Bites by triatomine vectors still represent the main route of transmission of *T. cruzi* to humans, but transmission is also possible orally (i.e., by ingestion of contaminated juices and foods) [3], congenitally from mother to child, by blood transfusion or organ donations [3–8].

In less than 5% of cases, individuals develop an acute symptomatic form of the disease [4]. After the acute phase is over, and if the infection remains untreated, *T. cruzi* progressively sets in the tissues with a high tropism for the heart and digestive tract. However, 60–70% of infected patients will not ever develop clinically significant organ damage [4]. Cardiac involvement is the most common organ damage, known as Chagasic cardiomyopathy (CC), which accounts for 25–30% of patients suffering from *T. cruzi* infection. Conduction disorders and dilatation of cardiac structures are typical manifestations, whose clinical spectrum comprise from asymptomatic to fatal outcomes [4]. Digestive involvement of CD (DCD) is less common (10–20%) and includes dilatation and dysfunction of digestive structures -mainly esophagus and colon [4].

Serological tests are the technique of choice for diagnosis of chronic *T. cruzi* infection [5]. Screening for cardiac involvement is indicated even in asymptomatic patients. The 12 lead ECG is the recommended test to detect conduction disorders and arrhythmias and should be offered to all patients [6,7]. Structural cardiomyopathy is classically assessed by thorax radiography, but last consensus recommends echocardiogram as it has more sensitivity and specificity [7–9]. Barium enema and esophagogram are recommended to assess digestive abnormalities due to CD but should only be performed when digestive symptoms are present [10].

In the last two decades, the global distribution of *T. cruzi* infection has changed due to population migration from endemic Latin America countries to Europe and other non-endemic settings [11]. In the early 2000s there was a large migration of Bolivian population towards Europe, mainly to large urban areas in Spain, due to the economic crisis and immigration restrictions in place in the USA [12].

A 2015 metanalysis estimated an average prevalence of *T. cruzi* infection of 4.2% in Latin American migrants living in European countries and, currently, Spain is the European country with the highest estimated prevalence [13]. Moreover, new cases of *T. cruzi* infection can occur in these traditionally non-endemic countries by non-vector-borne transmission. Thus, although screening for *T. cruzi* in populations originating from high endemic areas in Latin America should recommended in non-endemic countries, most European countries do not include this approach in their national health policies [14]. In Spain, there is no common legislation or initiative for *T. cruzi* infection has been implemented since 2010 [15].

In the last twenty years, several descriptions about clinical and epidemiological features of migrants at risk of *T. cruzi* infection have been published, but usually focused on blood donors or pregnant women, and the sample sizes of such studies were not large enough to describe the epidemiological profile of this population [16–20]. Here we present the clinical and epidemiological profile of a large sample of Latin American migrants attending the International Health Service (IHS) of the Hospital Clínic of Barcelona, Spain, which is the reference center for tropical diseases in the region.

Materials and methods

Ethics statement

All participants provided written informed consent for screening and data collection, in compliance with all bioethical regulations. All participants were over 16 years of age. Thus, none of the participants required consent by representation, following applicable legislation [21]. This study has been approved by the Ethics Committee named "Comité de Ética de la Investigación con medicamentos del Hospital Clínic de Barcelona" (reference HCB/2021/0174).

Design, setting, and recruitment

Barcelona has been one of the urban areas in Spain where a large Bolivian migrant population settled in the early 2000s [12], and the IHS of Hospital Clinic de Barcelona and Barcelona Institute of Global Health (ISGlobal) has a 17-year of experience in the clinical management and follow-up of this population. We conducted a descriptive retrospective study of 2,820 individuals originating from endemic Latin American countries considered *T. cruzi* endemic areas. Study subjects attended the medical consultation outpatient clinic of the IHS from March 2002 to March 2019. Subjects came to IHS (i) voluntarily, (ii) for screening after a positive result of a relative, (iii) upon advice by family or friends, (iv) referred by another doctor and (v) following information, education, and communication (IEC) community outreach.

The following countries were considered *T. cruzi* endemic areas: Argentina, Belize, Bolivia, Brazil, Chile, Colombia, Costa Rica, Ecuador, El Salvador, Guatemala, French Guyana, Guyana, Honduras, Mexico, Nicaragua, Panama, Paraguay, Peru, Suriname, Uruguay and Venezuela. Exclusion criteria for individuals to be included in the cohort were: a) individuals born in a non-endemic country; b) individuals under 16-years-old; c) inconclusive serology test; d) individuals who had not signed the informed consent.

Epidemiological and clinical data were collected from electronic medical records and questionnaires over the study period and registered in a database [22], and published in a data repository [23]. Collected variables included age, sex, country and department of origin, date of arrival to Spain, risk factors for *T. cruzi* infection and *T. cruzi* serology result. Also, the results of complementary tests performed by protocol were collected: ECG and echocardiogram, and barium enema and esophagogram when performed. Patients were then classified as follows: *T. cruzi* infection with no evidence of organ damage, CC and DCD. Only the first evaluation was included in this study to ascertain the phase of the infection at diagnosis.

Diagnosis of T. cruzi infection and organ damage

Diagnosis of *T. cruzi* infection was performed by two serological tests using different antigens, two different enzyme-linked immunosorbent assays (ELISA) methods, ORTHO Trypanosoma cruzi ELISA Test System (Johnson and Johnson, High Wycombe, United Kingdom) and Bioelisa Chagas (Biokit, Lliçà d' Amunt, Spain) during the period (2002–2012) and chemiluminescent microparticle assay (CMIA ARCHITECT Chagas Abbot) and ELISA (Chagas-ELISA Vircell, Spain) in the period (2013–2019). Indirect Immunofluorescence (IFA Vircell, Spain) and western blot were used in case of discordant serologic results [24].

Patients with two positive tests were classified as suffering from *T. cruzi* infection. CC was assessed through clinical history, physical exam, a 12-lead ECG, a thorax radiography, and echocardiogram. The following ECG abnormalities were considered as CC: bradycardia less than 50 beats per minute, auriculoventricular block (AVB), any type of bundle branch block or hemiblock, frequent ventricular premature beats, tachyarrhythmia of any origin, ST-T changes, electrically inactive areas, and pacemaker rhythm as it implies a conduction disorder [18]. The following echocardiogram abnormalities were considered as CC: any ventricular or atrial dilatation, contractility disorders, wall aneurisms and diastolic dysfunction [25,26]. Barium enema or esophagogram were performed when lower or upper digestive symptoms (respectively) compatible with CD were present. Colon dilatation in the barium enema was consider as DCD. Megacolon was defined as a diameter of more than 6.5 centimetres in

descending colon, more than 8 centimetres in ascending colon and more than 12 centimetres in cecum [27,28]. Any gastroesophageal dysfunction or esophageal dilatation was also considered as DCD. Only the first evaluation for *T. cruzi* infection for each patient was included in this study.

Statistical analysis

Data were described as frequencies, means (standard deviation, SD) or medians (interquartile range, IQR) for qualitative and quantitative variables, respectively. Chi-squared and Wilcoxon Rank Sum test were performed to assess differences between groups for qualitative and quantitative variables, respectively. Venn diagrams were used to summarize the relationship between *T. cruzi* infection risk factors. To quantify the relationship between patient characteristics and *T. cruzi* infection, odds ratios (OR) were estimated by means of logistic regression models. The significance level was set at 0.05. The analysis was carried out using Stata17 (@StataCorp. 2019. Stata Statistical Software: Release 16. College Station, TX: StataCorp LLC). For graphical representation R Foundation for Statistical Computing (@R Core Team. 2019. R: A Language and Environment for Statistical Computing. R Foundation for Statistical Computing. Vienna, Austria), with the package venneuler (Wilkinson, L. 2011. Venneuler: Venn and Euler Diagrams. R package version 1.1–0), were employed.

Results

General demographic and epidemiological variables

A total of 2,820 individuals born in *T. cruzi* endemic countries were included in the 17 years study period. The number of patients tested for *T. cruzi* infection per year progressively increased until 2009, when there was a peak of 356 individuals. The mean age was 36.8 years (SD 11.0). The median (IQR) of the duration of residence in Spain until the first visit to the IHS was six years (3.00-10.00) [2,813]. Most individuals came from Bolivia (Table 1), and 93.2% (n = 2,275/2,441) of these where from Cochabamba, Santa Cruz, and Chuquisaca Departments, all of which are considered highly endemic areas for *T. cruzi* transmission (Table 1). The overlap between reported risk factors is represented in the Fig 1. The greatest overlap (n = 987/1,889 individuals, 52.2%) was between the three risk factors related to the vector (rural area, adobe house, and exposure to the vector), while 101 individuals (N = 1,889; 5.35%) reported also having a mother with *T. cruzi* infection, and 114 (N = 1,889; 6.0%) had been residing in a rural endemic area and in an adobe house.

Seropositivity and its associated factors

Among 2,820 participants, 1,382 participants were diagnosed of *T. cruzi* infection based on serology (Fig 2). Variables associated with *T. cruzi* infection were analyzed and are summarized in Table 2. Proportion of *T. cruzi* infection by country is detailed in Table 3.

Organ damage in patients with T. cruzi infection: Chronic Chagas disease

Out of 1,382 patients with a confirmed positive result for *T. cruzi*, the first ECG of 662 patients was recorded in the study database. Among them, 79.0% (n = 523/662) had a normal result, while 21.0% (n = 139/662) had an altered result and 22 of them were found to have more than one disorder in the ECG. Among the 178 patients with an echocardiogram performed and recorded in the database, a total of 122 patients (68.5%) had a normal result, while 56 of them (31.5%) had an altered result. Cardiac dilatation was the most frequent disorder (n = 19; 10.67%): 12 of them with ventricular dilatation (6.7%), seven patients with atrial dilatation

Variable	n	Percentage (%)
Female sex (N = 2,820)	1,993	70.7
Age range ¹ (N = 2,820)		
<18	93	3.3
18-34	1,175	41.7
35-49	1,173	41.6
50-65	356	12.6
<65	23	0.8
Country of origin (N = 2,820)		
Bolivia	2,441	86.6
Cochabamba department	991	40.8
Santa Cruz department	783	32.2
Chuquisaca department	501	21.0
Other departments	155	6.4
Unknown	9	0.4
Ecuador	80	2.8
Argentina	73	2.6
Colombia	51	1.8
Peru	46	1.6
Paraguay	34	1.2
Brazil	24	0.9
Honduras	19	0.7
Venezuela	16	0.6
El Salvador	15	0.5
Chile	6	0.2
Mexico	6	0.2
Uruguay	4	0.1
Guatemala	3	0.1
Nicaragua	2	0.1
Referral (N = 1,893)		
Advised by family member or friend	1,081	57.1
Referred by a doctor ²	495	26.2
IEC ³	129	6.8
Own initiative	37	2.0
Other	150	7.9
Vectorial risk factors (N = 1,889)		
Rural area	1,517	80.3
Adobe house	1,506	79.7
Contact with vector	1,517	81.5
Non-vectorial risk factors		
Blood transfusion receptor (N = 2,773)	188	6.7
Mother with <i>T. cruzi</i> infection ($N = 2,746$)	232	8.5
Report a previous test before this assessment (N = 1,916)		
Yes	657	34.3
No	1,259	65.7
Positive serology (N = 2,820)	1,382	49.0

Table 1. General demographic and epidemiological variables of participants (N = 2,820).

1. Age range is expressed in years.

2. Among individuals referred by a doctor, 73.5% (n = 364/495) were referred by a primary care doctor.

3. IEC: Information, education, and communication.

https://doi.org/10.1371/journal.pntd.0011330.t001





https://doi.org/10.1371/journal.pntd.0011330.g001

(3.9%); two of them having a four-chamber dilatation (1.1%). Other findings, not considered to be CD related were: 14 patients with mild valve insufficiency, four of them with more than one valvopathy (11 cases of tricuspid insufficiency, six of mitral insufficiency, and one case of aortic insufficiency), and six had left ventricular hypertrophy. One patient had a severe pericardial effusion due to a non-chagasic etiology. Out of 153 patients with registered ejection fraction, six had a left ventricle ejection fraction (LVEF) of less than 35%, two patients had a LVEF between 35 and 45% and 145 had more than 45%. Main findings in the ECG and echocardiogram are summarized in Table 4.

In sum, 148 participants were diagnosed with CC. Of them, 16 patients (10.8%) had a normal ECG but with altered echocardiogram, nine of whom due to diastolic dysfunction (56.3% of them).

An esophagogram was recorded in 66 patients. Of these, 89.4% had a normal test (n = 59) and seven patients had a gastroesophageal dysfunction. Barium enema was registered in 71 patients, of whom 30 (42.3%) had a normal test. Dolichocolon was present in 29 patients, colon dilatation in 11 patients, including two who met the megacolon criteria. Cardiac involvement was present in patients from Argentina (n = 2), Bolivia (n = 138), Brazil (n = 3), Colombia (n = 2), Ecuador (n = 1), Paraguay (n = 1) and Venezuela (n = 1), while it was not present



Fig 2. Flowchart of screened individuals. Flowchart of screened participants of this study. Diagnosis of *T. cruzi* infection was based on the serology.

https://doi.org/10.1371/journal.pntd.0011330.g002

in patients from other countries. Patients with digestive involvement were from Argentina (n = 1) and Bolivia (n = 22), while it was not present in patients from other countries. Regarding the frequency of cardiac or digestive involvement in the first evaluation, 148 patients (17.1%) were registered to have CC, while 23 patients (2.6%) were recorded as DCD and 703 patients (80.8%) did not have any evidence of organ damage. Due to missing data, 512 patients were excluded of this analysis.

Discussion

Several studies performed in non-endemic countries have assessed the clinical and epidemiological profile of people at risk for *T. cruzi* infection [13,17-19,29,30]. However, most of these studies, especially in the USA [31-33], had either small sample sizes or focused on specific populations (e.g., blood donors or pregnant women). Furthermore, prevalence of *T. cruzi* infection in these studies was very low because most of the migrants who settled in North America originated from countries with low *T. cruzi* prevalence [31-33]. To our knowledge, this study describes one of the largest sample of patients at risk of *T. cruzi* infection in a nonendemic setting.

Trends in the influx of medical consultations are determined by Bolivian population migratory flows, since this population accounts for 86.6% of the participants. It is worth noting that patients from our study spent a median of six years in Spain before screening for *T. cruzi* infection was performed, which is longer than observed in other previous European studies [29,34].

Variable		Positive serology n/N (%)	p-value ¹	Adjusted OR (95% CI)	p-value ²
Sex	Male	236 / 566 (41.7)	0.0091	1	0.0454
	Female	638 / 1323 (48.2)		1.24 (1.00, 1.54)	
Age in years	<18	3 / 65 (4.6)	< 0.0001	1	< 0.0001
	18-34	313 / 733 (42.7)		10.07 (3.08, 32.90)	
	35-49	401 / 809 (49.6)		12.84 (3.93, 41.94)	
	50-65	150 / 267 (56.2)		19.50 (5.86, 64.93)	
	>65	7 / 15 (46.7)		15.34 (3.16, 74.56)	
Born in Bolivia	Yes	833/1,667 (50.0)	< 0.0001	3.12 (2.14, 546)	
	No	41/181 (18.5)		1	< 0.0001
Resided in rural area	Resided in rural area Yes 768/1,5		< 0.0001	1.56 (1.15, 2.11)	0.0039
	No	106/372 (28.5)		1	
Resided in adobe house	adobe house Yes 775/1,506 (51.5)		< 0.0001	1.73 (1.27, 2.35)	0.0004
	No	99/383 (25.9)		1	
Contact with vector	Yes	795/1,540 (51.6)	< 0.0001	1.97 (1.46, 2.65)	< 0.0001
	No	79/349 (22.6)		1	
	Total	874/1,889 (46.3)			

Table 2. Variables associated with T. cruzi infection using a multivariable model.

1: Chi-squared test

2: Wald test

OR: Odds ratio, CI: confidence interval

Note that we are summarizing only those observations with complete data for the multivariable model (N = 1,889)

https://doi.org/10.1371/journal.pntd.0011330.t002

This delay may be explained by barriers to healthcare, such as the population's legal immigration and residential status, poor socio-economic conditions, and cultural barriers. It is estimated that a significant number of people living in Spain with *T. cruzi* infection are unaware of their condition. In fact, the index of underdiagnosed cases of *T. cruzi* infection is estimated to be of 71% [35] and, indeed, most of our participants had never been tested for *T. cruzi*

Table 3.	Frequency of	T. cruzi	infection	according t	o country	of origin
----------	--------------	----------	-----------	-------------	-----------	-----------

Country of origin	Positive for T. cruzi test
Bolivia (n = 2,441)	1,304 (53.4%)
Paraguay (n = 34)	17 (50.0%)
Argentina (n = 73)	31 (42.5%)
Brazil (n = 24)	8 (33.3%)
Chile $(n = 6)$	2 (33.3%)
Venezuela (n = 16)	4 (25.0%)
El Salvador (n = 15)	3 (20.0%)
Mexico (n = 6)	1 (16.7%)
Honduras (n = 19)	2 (10.5%)
Ecuador (n = 80)	6 (7.5%)
Peru (n = 46)	3 (6.5%)
Colombia (n = 51)	3 (5.9%)
Guatemala (n = 3)	0 (0.0%)
Nicaragua (n = 2)	0 (0.0%)
Uruguay (n = 4)	0 (0.0%)

https://doi.org/10.1371/journal.pntd.0011330.t003

ECG findings	Frequency	Proportion among patients with altered ECG (n = 139)	Proportion among patients with ECG performed (n = 662)
Normal ECG	523	-	79.0%
RBBB	64	46.0%	9.7%
LAFB	40	28.8%	6.0%
Sinus bradycardia	17	12.2%	2.6%
RBBB + LAFB	13	9.4%	2.0%
AVB	11	7.9%	1.7%
Nonspecific ST-T changes	7	5.0%	1.1%
VES	6	4.3%	0.9%
Supraventricular arrythms	6	4.3%	0.9%
Pacemaker	5	3.6%	0.8%
Electrically inactive area	2	1.4%	0.3%
Low voltage	1	0.7%	0.2%
LBBB	1	0.7%	0.2%
Other disorders	5	3.6%	0.8%
Echocardiogram findings	Frequency	Proportion among patients with altered echocardiogram $(n = 56)$	Proportion among patients with echocardiogram performed $(n = 178)$
Normal echocardiogram	122	-	68.6%
Mild valve insufficiency	14	25.1%	7.9%
Ventricular dilatation	12	21.4%	6.7%
Diastolic dysfunction	10	17.9%	5.6%
Atrial dilatation	7	12.5%	3.9%
Contractility disorders	7	12.5%	3.9%
Ventricular hypertrophy	5	8.9%	2.8%
Wall aneurysm	4	7.1%	2.3%
Global hypokinesia	4	7.1%	2.3%
Other disorders	4	7.1%	2.%

Table 4.	Main ECG and	l echocardiogram	findings in	patients with	T. cruzi infection.
			· · · ·		

RBBB. Right bundle branch block. LAFB: Left anterior fascicular block. AVB: Atrioventricular block. VES: Ventricular extrasystoles. LBBB: Left bundle branch block. LPFB: Left posterior fascicular block.

https://doi.org/10.1371/journal.pntd.0011330.t004

infection prior to our assessment. The great predominance of women and working-age people from endemic countries reflects the demographic profile of Latin-American migrants who came to Europe in the early 2000's, as job offers were usually as domestic workers, which was socially attributed to women [12].

Health promotion campaigns among the Bolivian community were decisive in defining the attendance trends in the medical attention seeking in the IHS of Hospital Clinic de Barcelona, as shown by several health promotion programs directed to communities from *T. cruzi* endemic countries [36]. Most patients came because they were advised by a family member or a friend, which reveals the great importance of the social capital in the knowledge and care of the disease.

This study presents higher proportion of *T. cruzi* infection among participants than other European studies [13]. However, a selection bias was present because the participants were attending a referral center to which sometimes they were directed after a positive test. For this reason, proportion of seropositivity from this study should never be used to estimate the prevalence of *T. cruzi* infection in the general Latin American population living in Spain, which was already estimated to be of 2.1% in a recent study [35]. Nevertheless, we must consider that

Latin Americans in Barcelona come from highly endemic areas compared with those in other European cities [37]. In fact, most individuals were from the Bolivian departments of Cochabamba, Santa Cruz, and Chuquisaca, which are hyperendemic areas of *T. cruzi* infection [5].

Factors of potential vector exposure (adobe house, rural area, to have been in contact with the vector) were present in most patients, but there was a significant overlap among all the risk factors. Thus, many participants who had received a transfusion in an endemic country or whose mother had *T. cruzi* infection, have also had potential vector exposure.

Congenital and transfusion risks account for the possibility of new cases in Europe and justifies the implementation of screening programs for these risk groups as a priority in health public policies [14]. Moreover, screening for *T. cruzi* infection in all Latin American adults living in Europe has been shown to be cost-effective [37]. Additionally, young people infected with *T. cruzi* on an early chronic phase of *T. cruzi* infection seems to be the most benefited from an early diagnosis and anti-trypanosoma therapy [38,39]. Even more so in women of childbearing age, since treatment before pregnancy dramatically reduces the chances of congenital transmission [40,41].

Results from ECG assessment show RBBB as the most frequent and with the strongest association with T. cruzi infection, consistent with previously published studies [42]. LAFB was more frequent than sinus bradycardia, contrary to what was observed in previous studies [20,43]. Although the sensitivity of the ECG may be limited because it cannot evaluate function and morphology, it is an accessible and low-cost test capable of detecting potentially serious abnormalities. Indeed, more than 10% of patients with diagnosis of cardiac involvement had a normal ECG result and were diagnosed by echocardiogram. Similar data observing altered echocardiogram with normal ECG have been previously published [44,45]. Therefore, echocardiogram is considered a necessary test for this evaluation by most authors and guidelines [7–9], our findings pointing in the same direction. Typically, wall dilatation has been considered the main echocardiogram disorder in patients with T. cruzi infection, but diastolic dysfunction is suggested to be an earlier alteration of CC [26,46]. Our study shows diastolic dysfunction as the second most frequent disorder among patients with T. cruzi infection. Although this frequency is not very high compared with that of general population [47], its presence may be considered of Chagasic etiology, especially taking into account the young age of our patients. Presence of isolated mild tricuspid or mitral insufficiency without dilated ring should not be interpreted as a CC, as several studies have demonstrated a high prevalence (up to 80%) of mild regurgitation in healthy subjects [48] and, to our knowledge, there are no studies which associate it with disease progression. Regarding digestive involvement, dolichocolon was very frequent in patients with a barium enema performed. In agreement with previous work [10,49,50], dolichocolon was not considered as a digestive involvement for CD, thus the observed frequency of digestive involvement was relatively low. Furthermore, digestive involvement was only present in patients from southern countries of Latin America (Bolivia and Argentina). This would be consistent with the notion of tissue specific tropism of different strains of *T. cruzi* with characteristic geographic distribution [51], but the validity of this observation cannot be concluded since practically all the patients come from those same countries.

More than 80% of patients were not found to have CC or DCD in the first evaluation. This form of the disease is classically known as "indeterminate form" of chronic *T. cruzi* infection [4] or rather "chronic phase without demonstrable pathology" [52]. Most patients were young, so a number of them will develop some form of organ damage in the following years at a rate of 1–2% per year [53]. Hence, age could explain the lower prevalence of CC and DCD in studies from non-endemic countries compared with those from endemic countries [29]. The lack of clinical and biological markers of progression limits a more accurate follow-up in this big

group of patients. The future availability of such tools should be highly relevant for patients' clinical management and healthcare routes [54].

One limitation of this study is that comorbidities were not specifically addressed. We acknowledge that screening for comorbidities will become increasingly important with the aging of Latin American migrants, but we posit that the impact on the overall findings of this study would probably be small due to the mean and median age of the participants included in our study. Also recall bias could be present in the answers to the epidemiological questionnaires. As the participants were recruited in a regional reference center a selection bias was present. Observer bias could also be present as electrocardiogram is an observer-dependent test, and the questionnaires were filled in by different physicians. This bias was probably reduced as diagnostic criteria of electrocardiographic disorders were established in detail. We were able to include the status of the patient at the moment of the screening and at the moment of the initial diagnosis of T. cruzi infection, but not their follow-up, that is why the number of ECG and echocardiogram recorded for the analyses was little compared with total sample size. Despite this discordance, the absolute number of tests registered and analyzed was also high and allowed us to extract enough data for the analysis and discussion. It would be interesting to analyze the longitudinal evolution of this cohort in terms of the prognostic value of the alterations found in this first evaluation.

Conclusions

Currently, *T. cruzi* infection is distributed throughout Europe due to migratory flows from Latin America of the last two decades. Big Spanish cities have received the largest population at risk. Although arrivals from endemic countries have decreased in the last years, we still observe many individuals at risk who have not been previously tested. Moreover, since universal testing of pregnant women is not widely implemented, children with *T. cruzi* infection may be born and remain undiagnosed. In this study, individuals from endemic countries settled in Barcelona were mostly working-age women who came between 2004 and 2007. Migrants with a similar epidemiological profile, especially of Bolivian origin or descent, should be the target for a national or European screening program. Thus, health promotion programs and an adequate information to the patient in the consultation are essential to carry out the screening program.

In concurrence with other studies, previous contact with the vector was the most frequent and strongly associated risk factor with *T. cruzi* infection among those related with vector exposure.

The primary assessment in patients with chronic *T. cruzi* infection should include an ECG and echocardiogram. RBBB and LAFB were the most typical ECG disorders among *T. cruzi* infection patients in our series, while wall dilatation and diastolic dysfunction were the most frequent findings in the echocardiogram. In our series, as well as in other previous studies, a relevant proportion of patients with CC were diagnosed by echocardiogram having a normal ECG, which reinforces the echocardiogram as an essential test in the assessment of CC. Although dolichocolon was the most frequent finding in barium enema, it should not be interpreted as a digestive involvement of CD. It is expected that prevalence of CC and DCD will gradually increase in Europe as migrant populations settled in these countries age.

We are gradually learning more about *T. cruzi* infection, but there are still important knowledge gaps regarding this infection, especially about the lack of clinical or biological prognostic factors. Further studies will be necessary for a better understanding of its epidemiological and clinical behaviour.

Supporting information

S1 Table. The Supporting Information file contains the numbers used for the generation of Fig 1. The file consists of five columns, one for each risk factor, and a sixth column indicating the number of subjects in the corresponding situation. For each risk factor, its presence or absence is indicated by TRUE or FALSE, respectively. (CSV)

Acknowledgments

We first would like to thank all the participants of this study. We also thank the Hospital Clínic de Barcelona and ISGlobal, as well as the nursing team, the medical team, and the staff who have helped in this project.

Author Contributions

- **Conceptualization:** Pedro Laynez-Roldán, Irene Losada-Galván, Elizabeth Posada, Leonardo de la Torre Ávila, Inés Oliveira-Souto, Antonia Calvo-Cano, José Muñoz, Joaquim Gascón, Maria-Jesus Pinazo.
- Data curation: Pedro Laynez-Roldán, Irene Losada-Galván, Aina Casellas, Sergi Sanz, Carme Subirà, Isabel Vera, Montserrat Roldán, Edelweiss Aldasoro, Antonia Calvo-Cano, Maria-Eugenia Valls, Míriam J. Álvarez-Martínez, Alba Abras, Cristina Ballart, Maria-Jesus Pinazo.
- **Formal analysis:** Pedro Laynez-Roldán, Irene Losada-Galván, Aina Casellas, Sergi Sanz, Maria-Jesus Pinazo.
- Funding acquisition: Irene Losada-Galván, Joaquim Gascón, Maria-Jesus Pinazo.
- Investigation: Pedro Laynez-Roldán, Irene Losada-Galván, Elizabeth Posada, Leonardo de la Torre Ávila, Aina Casellas, Sergi Sanz, Natalia Rodriguez-Valero, Daniel Camprubí-Ferrer, Isabel Vera, Montserrat Roldán, Inés Oliveira-Souto, Antonia Calvo-Cano, Maria-Eugenia Valls, Míriam J. Álvarez-Martínez, Montserrat Gállego, Alba Abras, Cristina Ballart, José Muñoz, Joaquim Gascón, Maria-Jesus Pinazo.
- Methodology: Pedro Laynez-Roldán, Irene Losada-Galván, Aina Casellas, Sergi Sanz, Joaquim Gascón, Maria-Jesus Pinazo.
- Software: Aina Casellas, Sergi Sanz, Carme Subirà.
- Supervision: Irene Losada-Galván, Elizabeth Posada, Leonardo de la Torre Ávila, Natalia Rodriguez-Valero, Daniel Camprubí-Ferrer, Edelweiss Aldasoro, Inés Oliveira-Souto, Antonia Calvo-Cano, Montserrat Gállego, José Muñoz, Joaquim Gascón, Maria-Jesus Pinazo.
- Validation: Joaquim Gascón, Maria-Jesus Pinazo.
- Visualization: Pedro Laynez-Roldán, Irene Losada-Galván, Elizabeth Posada, Aina Casellas, Sergi Sanz, Carme Subirà.
- Writing original draft: Pedro Laynez-Roldán, Irene Losada-Galván, Joaquim Gascón, Maria-Jesus Pinazo.
- Writing review & editing: Pedro Laynez-Roldán, Irene Losada-Galván, Elizabeth Posada, Leonardo de la Torre Ávila, Aina Casellas, Sergi Sanz, Carme Subirà, Natalia Rodriguez-Valero, Daniel Camprubí-Ferrer, Isabel Vera, Montserrat Roldán, Edelweiss

Aldasoro, Inés Oliveira-Souto, Antonia Calvo-Cano, Maria-Eugenia Valls, Míriam J. Álvarez-Martínez, Montserrat Gállego, Alba Abras, Cristina Ballart, José Muñoz, Joaquim Gascón, Maria-Jesus Pinazo.

References

- Chagas C. Nova tripanozomiaze humana: estudos sobre a morfolojia e o ciclo evolutivo do Schizotrypanum cruzi n. gen., n. sp., ajente etiolojico de nova entidade morbida do homem. Mem Inst Oswaldo Cruz. 1909; 1: 159–218. https://doi.org/10.1590/s0074-02761909000200008
- 2. Integrating neglected tropical diseases into global health and development: fourth WHO report on neglected tropical diseases. World Health Organization. Geneva; 2017. Available: http://apps.who.int/iris/bitstream/10665/255011/1/9789241565448-eng.pdf?ua=1
- de Noya BA, Díaz-Bello Z, Colmenares C, Ruiz-Guevara R, Mauriello L, Muñoz-Calderón A, et al. Update on oral chagas disease outbreaks in Venezuela: Epidemiological, clinical and diagnostic approaches. Mem Inst Oswaldo Cruz. 2015; 110: 377–386. <u>https://doi.org/10.1590/0074-02760140285</u> PMID: 25946155
- Rassi A Jr, Marin-Neto A. Seminar Chagas disease. Lancet. 2010; 375: 1388–402. https://doi.org/10. 1016/S0140-6736(10)60061-X PMID: 20399979
- WHO Expert Committee. Control of Chagas disease. World Health Organ Tech Rep Ser. 2002; 905: i– vi, 1–109, back cover. Available: <u>http://www.ncbi.nlm.nih.gov/pubmed/12092045</u> PMID: <u>12092045</u>
- Kuschnir E, Sgammini H, Castro R, Eveguoz C, Ledesma R, Brunetto J. Evaluation of cardiac function by radioisotopic angiography, in patients with chronic Chagas cardiopathy. Arq Bras Cardiol. 1985; 45: 249–56. Available: http://enfermedadchagas.com.ar/clasif.pdf
- 7. Pinto Dias JC, Novaes Ramos A Jr, Dias Gontijo E, Luquetti A, Shikanai-Yasuda MA, Rodrigues Coura J, et al. 2nd Brazilian Consensus on Chagas disease, 2015. Rev Soc Bras Med Trop. 2016; 49: 3–60. https://doi.org/10.1590/0037-8682-0505-2016 PMID: 27982292
- Andrade JP, Marin-Neto JA, Paola AAV, Vilas-Boas F, Oliveira GMM, Bacal F, et al. Sociedade Brasileira de Cardiologia. I Diretriz Latino Americana para o Diagnóstico e Tratamiento da Cardiopatia Chagásica. Arq Bras Cardiol. 2011; 97: 1–48.
- Nunes MCP, Beaton A, Acquatella H, Bern C, Bolger AF, Echeverría LE, et al. Chagas Cardiomyopathy: An Update of Current Clinical Knowledge and Management: A Scientific Statement From the American Heart Association. Circulation. 2018. <u>https://doi.org/10.1161/CIR.000000000000599</u> PMID: 30354432
- Pinazo MJ, Cañas EG, Elizalde JI, García M, Gascón J, Gimeno F, et al. Diagnosis, management and treatment of chronic Chagas' gastrointestinal disease in areas where Trypanosoma cruzi infection is not endemic. Gastroenterol Hepatol. 2010; 33: 191–200. https://doi.org/10.1016/j.gastrohep.2009.07.009 PMID: 19837482
- Lidani KCF, Andrade FA, Bavia L, Damasceno FS, Beltrame MH, Messias-Reason IJ, et al. Chagas Disease: From Discovery to a Worldwide Health Problem. Front Public Health. 2019; 7: 1–13. https:// doi.org/10.3389/fpubh.2019.00166 PMID: 31312626
- 12. de la Torre Ávila L. No llores prenda, pronto volveré: Migración, movilidad social, herida familiar y desarrollo. La Paz: Institut français d'études andines; 2004. https://doi.org/10.4000/books.ifea.5303
- Requena-Méndez A, Aldasoro E, de Lazzari E, Sicuri E, Brown M, Moore DAJ, et al. Prevalence of Chagas Disease in Latin-American Migrants Living in Europe: A Systematic Review and Meta-analysis. PLoS Negl Trop Dis. 2015; 9: 1–15. https://doi.org/10.1371/journal.pntd.0003540 PMID: 25680190
- Requena-Méndez A, Albajar-Viñas P, Angheben A, Chiodini P, Gascón J, Muñoz J. Health Policies to Control Chagas Disease Transmission in European Countries. PLoS Negl Trop Dis. 2014; 8. https://doi. org/10.1371/journal.pntd.0003245 PMID: 25357193
- Autonomous government of Catalonia. Protocol for screening and diagnosing Chagas disease in pregnant Latin American women and their newborns. Departament de Salut Catalunya. 2018; 1: 53. Available: https://canalsalut.gencat.cat/web/.content/_A-Z/C/chagas/documents/arxius/ protoclcribratgeidiagnostic_angl.pdf
- Antinori S, Galimberti L, Grande R, Bianco R, Oreni L, Traversi L, et al. Chagas disease knocks on our door: a cross-sectional study among Latin American immigrants in Milan, Italy. Clinical Microbiology and Infection. 2018; 24: 1340.e1–1340.e6. https://doi.org/10.1016/j.cmi.2018.03.017 PMID: 29555394
- Piron M, Vergés M, Muñoz J, Casamitjana N, Sanz S, Maymó RM, et al. Seroprevalence of Trypanosoma cruzi infection in at-risk blood donors in Catalonia (Spain). Transfusion (Paris). 2008; 48: 1862– 1868. https://doi.org/10.1111/j.1537-2995.2008.01789.x PMID: 18522707

- Muñoz J, Prat JG i., Gállego M, Gimeno F, Treviño B, López-Chejade P, et al. Clinical profile of Trypanosoma cruzi infection in a non-endemic setting: Immigration and Chagas disease in Barcelona (Spain). Acta Trop. 2009; 111: 51–55. https://doi.org/10.1016/j.actatropica.2009.02.005 PMID: 19426663
- Barona-Vilar C, Giménez-Martí MJ, Fraile T, González-Steinbauer C, Parada C, Gil-Brusola A, et al. Prevalence of Trypanosoma cruzi infection in pregnant Latin American women and congenital transmission rate in a non-endemic area: The experience of the Valencian Health Programme (Spain). Epidemiol Infect. 2012; 140: 1896–1903. https://doi.org/10.1017/S0950268811002482 PMID: 22129521
- Gobbi F, Angheben A, Anselmi M, Postiglione C, Repetto E, Buonfrate D, et al. Profile of Trypanosoma cruzi Infection in a Tropical Medicine Reference Center, Northern Italy. PLoS Negl Trop Dis. 2014; 8: 5– 10. https://doi.org/10.1371/journal.pntd.0003361 PMID: 25502927
- Ley 41/2002, de 14 de noviembre, básica reguladora de la autonomía del paciente y de derechos y obligaciones en materia de información y documentación clínica. Boletín Oficial del Estado, núm 274, de 15 de noviembre de 2002. Available: https://www.boe.es/buscar/pdf/2002/BOE-A-2002-22188-consolidado.pdf
- Fundació Clinic-Gesem. 2009. Chagas Database. Barcelona; 2009. Available: <u>https://chagas-clinic.gesem.net/</u>
- Casellas Aina, Abras Alba, Calvo-Cano Antonia, Subirà Carme, Ballart Cristina, Camprubí-Ferrer Daniel, et al. {Epidemiological and clinical data of over 2800 individuals at risk for Trypanosoma cruzi infection in a non-endemic country. Repositori de Dades de Recerca; 2022. Available: https://doi.org/10.34810/data223n
- Riera C, Verges M, Iniesta L, Fisa R, Gállego M, Tebar S, et al. Short report: Identification of a western blot pattern for the specific diagnosis of Trypanosoma cruzi infection in human sera. American Journal of Tropical Medicine and Hygiene. 2012; 86: 412–416. https://doi.org/10.4269/ajtmh.2012.11–0111
- Barbosa MM, Nunes MCP. Risk stratification in chagas disease. Rev Esp Cardiol. 2012; 65: 17–21. https://doi.org/10.1016/j.recesp.2012.07.006 PMID: 22921170
- Acquatella H. Echocardiography in Chagas heart disease. Circulation. 2007; 115: 1124–1131. <u>https://doi.org/10.1161/CIRCULATIONAHA.106.627323 PMID: 17339570</u>
- Ximenes C, Rezende JM, Moreira H, Vaz M. Técnica simplificada para o diagnóstico radiológico do Megacolon chagásico. Rev Soc Bras Med Trop. 1984; 17–23.
- de Rezende J, Lauar KM, de Oliveira A. Clinical and radiological aspects of aperistalsis of the esophagus. Rev Bras Gastroenterol. 1960; 247–262. PMID: 13741121
- Salvador F, Treviño B, Sulleiro E, Pou D, Sánchez-Montalvá A, Cabezos J, et al. Trypanosoma cruzi infection in a non-endemic country: Epidemiological and clinical profile. Clinical Microbiology and Infection. 2014; 20: 706–712. https://doi.org/10.1111/1469-0691.12443 PMID: 24329884
- Beatty NL, Forsyth CJ, Burkett-Cadena N, Wisely SM. Our Current Understanding of Chagas Disease and Trypanosoma cruzi Infection in the State of Florida—an Update on Research in this Region of the USA. Current Tropical Medicine Reports. Springer Science and Business Media Deutschland GmbH; 2022. https://doi.org/10.1007/s40475-022-00261-w
- Leiby DA, Herron RM, Read EJ, Lenes BA, Stumpf RJ. Trypanosoma cruzi in Los Angeles and Miami blood donors: Impact of evolving donor demographics on seroprevalence and implications for transfusion transmission. Transfusion (Paris). 2002; 42: 549–555. https://doi.org/10.1046/j.1537-2995.2002. 00077.x PMID: 12084162
- Centers for Disease Control and Prevention (CDC). Blood Donor Screening for Chagas disease—US 2006–2007. 2007 [cited 1 Oct 2019] pp. 141–143. Available: <u>https://www.cdc.gov/mmwr/preview/</u> mmwrhtml/mm5607a2.htm
- Custer B, Agapova M, Bruhn R, Cusick R, Kamel H, Tomasulo P, et al. Epidemiologic and laboratory findings from 3 years of testing United States blood donors for Trypanosoma cruzi. Transfusion (Paris). 2012; 52: 1901–1911. https://doi.org/10.1111/j.1537-2995.2012.03569.x PMID: 22339233
- Jackson Y, Gétaz L, Wolff H, Holst M, Mauris A, Tardin A, et al. Prevalence, clinical staging and risk for blood-borne transmission of chagas disease among latin American migrants in Geneva, Switzerland. PLoS Negl Trop Dis. 2010; 4: 1–7. https://doi.org/10.1371/journal.pntd.0000592 PMID: 20126397
- Navarro M, Reguero L, Subirà C, Blázquez-Pérez A, Requena-Méndez A. Estimating chagas disease prevalence and number of underdiagnosed, and undertreated individuals in Spain. Travel Med Infect Dis. 2022; 47. https://doi.org/10.1016/j.tmaid.2022.102284 PMID: 35245657
- **36.** Sanmartino M, Avaria Saavedra A, de la Torre Ávila L. A Social Approach to Chagas Disease: A First Step to Improve Access to Comprehensive Care. Chagas Disease. 2020; 57–74. https://doi.org/10. 1007/978-3-030-44054-1_4

- Requena-Méndez A, Bussion S, Aldasoro E, Jackson Y, Angheben A, Moore D, et al. Cost-effectiveness of Chagas disease screening in Latin American migrants at primary health-care centres in Europe: a Markov model analysis. Lancet Glob Health. 2017; 5: e439–e447. https://doi.org/10.1016/S2214-109X(17)30073-6 PMID: 28256340
- Viotti R, Vigliano C, Lococo B, Bertocchi G, Petti M, Alvarez MG, et al. Long-term cardiac outcomes of treating chronic chagas disease with benznidazole versus no treatment: A nonrandomized trial. Ann Intern Med. 2006. https://doi.org/10.7326/0003-4819-144-10-200605160-00006 PMID: 16702588
- Morillo C.A., Marin-Neto J.A., Avezum A., Sosa-Estani S., Rassi A. FR Jr., et al. Randomized trial of benznidazole for chronic Chagas' Cardiomyopathy. The New England Journal of Medicine. 2015. pp. 1295–306. https://doi.org/10.1056/NEJMoa1507574 PMID: 26323937
- Moscatelli G, Moroni S, García-Bournissen F, Ballering G, Bisio M, Freilij H, et al. Prevention of congenital chagas through treatment of girls and women of childbearing age. Mem Inst Oswaldo Cruz. 2015; 110: 507–509. https://doi.org/10.1590/0074-02760140347 PMID: 25993401
- Fabbro DL, Danesi E, Olivera V, Codebó MO, Denner S, Heredia C, et al. Trypanocide Treatment of Women Infected with Trypanosoma cruzi and Its Effect on Preventing Congenital Chagas. PLoS Negl Trop Dis. 2014; 8. https://doi.org/10.1371/journal.pntd.0003312 PMID: 25411847
- Rojas LZ, Glisic M, Pletsch-Borba L, Echeverría LE, Bramer WM, Bano A, et al. Electrocardiographic abnormalities in Chagas disease in the general population: A systematic review and meta-analysis. PLoS Negl Trop Dis. 2018; 12: 1–20. https://doi.org/10.1371/journal.pntd.0006567 PMID: 29897909
- 43. Sánchez-Montalvá A, Salvador F, Rodríguez-Palomares J, Sulleiro E, Sao-Avilés A, Roure S, et al. Chagas cardiomyopathy: Usefulness of ekg and echocardiogram in a Non-Endemic country. PLoS One. 2016; 11: 1–13. https://doi.org/10.1371/journal.pone.0157597 PMID: 27308824
- Viotti RJ, Vigliano C, Laucella S, Lococo B, Petti M, Bertocchi G, et al. Value of echocardiography for diagnosis and prognosis of chronic Chagas disease cardiomyopathy without heart failure. Heart. 2004; 90: 655–660. https://doi.org/10.1136/hrt.2003.018960 PMID: 15145872
- 45. dos Reis Velloso Siciliano AP, Hasslocher-Moreno A, Silvestre de Sousa A, Emanuel Alvarenga Americano do Brasil P, Teixeira de Holanda M, Yorio Garcia M, et al. Padrão da Função Diastólica e Relação com a Gravidade da Forma Clínica em 902 Pacientes na Fase Crônica da Doença de Chagas Pattern of Diastolic Function and its Relation to the Severity of the Clinical Form in 902 Patients in the Chronic Phase of Chagas Disease. Revista da SOCERJ. 2006;N. Available: http://sociedades.cardiol.br/socerj/ revista/2006_01/a2006_v19_n01_art10.pdf
- 46. Refoyo E, Troya J, Trigo E, Guzmán-Martinez G, Valbuena-López S, Caro-Codón J, et al. Comparison of noninvasive cardiac test strategies for newly diagnosed chagas disease in a non-endemic zone. American Journal of Tropical Medicine and Hygiene. 2020; 103: 1480–1486. https://doi.org/10.4269/ajtmh.20-0389 PMID: 32700660
- Kuznetsova T, Herbots L, López B, Jin Y, Richart T, Thijs L, et al. Prevalence of left ventricular diastolic dysfunction in a general population. Circ Heart Fail. 2009; 2: 105–112. https://doi.org/10.1161/ CIRCHEARTFAILURE.108.822627 PMID: 19808325
- Okura H, Takada Y, Yamabe A, Ozaki T, Yamagishi H, Toda I, et al. Prevalence and correlates of physiological valvular regurgitation in healthy subjects: A color Doppler echocardiographic study in the current era. Circulation Journal. 2011; 75: 2699–2704. https://doi.org/10.1253/circj.CJ-11-0277 PMID: 21881247
- 49. Pinto JJ, Pinazo MJ, Saravia J, Gainsborg I, Magne HR, Cuatrecasas M, et al. Characterization of digestive disorders of patients with chronic Chagas disease in Cochabamba, Bolivia. Heliyon. 2019; 5: 1–12. https://doi.org/10.1016/j.heliyon.2019.e01206 PMID: 30788442
- Pinazo MJ, Lacima G, Elizalde JI, Posada EJ, Gimeno F, Aldasoro E, et al. Characterization of Digestive Involvement in Patients with Chronic T. cruzi Infection in Barcelona, Spain. PLoS Negl Trop Dis. 2014; 8: 1–7. https://doi.org/10.1371/journal.pntd.0003105 PMID: 25144648
- Higo H, Miura S, Horio M, Mimori T, Hamano S, Agatsuma T, et al. Genotypic variation among lineages of Trypanosoma cruzi and its geographic aspects. Parasitol Int. 2004; 53: 337–344. <u>https://doi.org/10.1016/j.parint.2004.06.001</u> PMID: 15464443
- 52. Comité Científico de Enfermedad de Chagas de La Federación Argentina de Cardiología 2009., Consejo de Miocardiopatías y Enfermedad de Chagas de la Sociedad Interamericana de Cardiología. Clasificación Enfermedad de Chagas. Consenso Internacional Buenos Aires 2010. (Buenos Aires, Argentina. Marzo 20 de 2010). Revista de la Federación Argentina de Cardiología. 2010.
- Murcia L, Carrilero B, Muñoz MJ, Iborra MA, Segovia M. Usefulness of PCR for monitoring benznidazole response in patients with chronic Chagas' disease: A prospective study in a non-disease-endemic country. Journal of Antimicrobial Chemotherapy. 2010; 65: 1759–1764. <u>https://doi.org/10.1093/jac/ dkq201 PMID: 20542903</u>

54. Cortes-Serra N, Losada-Galvan I, Pinazo M-J, Fernandez-Becerra C, Gascon J, Alonso-Padilla J. State-of-the-art in host-derived biomarkers of Chagas disease prognosis and early evaluation of anti-Trypanosoma cruzi treatment response. Biochim Biophys Acta Mol Basis Dis. 2020; 1866: 165758. https://doi.org/10.1016/j.bbadis.2020.165758 PMID: 32169507